Author's response to reviews

Title: Use of selective serotonin reuptake inhibitors and risk of re-operation due to post-surgical bleeding in breast cancer patients: a Danish population-based cohort study

Authors:

Rune Gärtner (rune.gartner@rh.regionh.dk)
Deirdre Cronin-Fenton (dc@dce.au.dk)
Heidi H. Hundborg (hh@dce.au.dk)
Lars Pedersen (lap@dce.au.dk)
Timothy L. Lash (tlash@bu.edu)
Henrik T. Sørensen (hts@dce.au.dk)
Niels Kroman (kroman@dadlnet.dk)

Version: 5 Date: 14 January 2010

Author's response to reviews: see over
Cover Letter

Corresponding/submitting author:
Henrik Toft Sørensen

Address: Department of Clinical Epidemiology, Aarhus University Hospital. Olof Palmes Alle 43-45, Aarhus N 8200, Denmark

Phone number: +45 8942 4800
FAX number: + 45 8942 4801

E-mail address: HTS@DCE.AU.DK

Aarhus 10.12.2009

Dear Ms. Norton,

We are very happy to hear that BMC surgery will publish our manuscript “Use of selective serotonin reuptake inhibitors and risk of re-operation due to post-surgical bleeding in breast cancer patients: a Danish population-based cohort study”.

Thank you for the last comments on the format of the manuscript. The requested major as well as minor revisions have been made. Please see the resubmitted manuscript, and below for a point-by-point description of the changes made.

Please do not hesitate to contact me if you have any further questions or requirements regarding the revised manuscript.

Yours sincerely,
Henrik Toft Sørensen
Corresponding/submitting author
Formatting changes to be made:
-----------------------------
Major revisions (we require the author to make these changes)

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].'
DONE

Tables: Every table provided at the end of the manuscript file must be referred to in the text. Currently we can not find table 2 cited
DONE – see results section

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efg@generic.co.uk
IJ: IJKL@corporation.com
DONE

Word count: please remove these
DONE

Underlining: please note that underlining is not supported, you may wish to change this to bold
DONE – changed to bold

Abstract structure: please rename the section Introduction to Background. Please also remember to update the abstract in the submission system
DONE

Keywords: please remove these
DONE

Introduction: please rename this section Background
DONE

Tables: the table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table
DONE

Tables- please remove the visible vertical lines from your tables
DONE

Appendix: please move this section to a position immediately after the authors’ contributions and ensure it is cited in the text
DONE

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
DONE – page 4 line 2, full stop deleted
page 5 line 4, space deleted, line 12 space inserted
page 6 last line, brackets corrected
page 10 line 12, space deleted
page 11 line 9, space deleted